Author's response to reviews

Title: Human papillomavirus-mediated carcinogenesis and HPV-associated oral and oropharyngeal squamous cell carcinoma. Part II: Human papillomavirus associated oral and oropharyngeal squamous cell carcinoma

Authors:

Liviu Feller (lfeller@ul.ac.za)
Neil H Wood (oralmed@ul.ac.za)
Razia AG Khammissa (rkhammissa@webmail.co.za)
Johan Lemmer (jbowman@iburst.co.za)

Version: 3 Date: 31 May 2010

Author's response to reviews:

The following corrections were made to the manuscript as requested:
1) The list of authors in the manuscript is written exactly as it is in the submission system, both in style and order.
2) The highlighting has been removed
3) On page 5, in the second paragraph, the words "squamous cell carcinoma" have been changed to "SCC".

We trust you will find these in order.